

Published online: 27 June 2015 © The Association of Bone and Joint Surgeons® 2015

Cochrane in CORR[®]: Interventions for Treating Proximal Humeral Fractures in Adults (Review)

Ydo Vincent Kleinlugtenbelt MD, Mohit Bhandari MD, PhD, FRCSC

Importance of the Topic

Proximal humeral fractures are common fractures that account for 6% of all adult fractures [4] with an overall incidence of 19 per 100,000 [8]. Neer estimated that

A Note from the Editor-In-Chief: We are pleased to publish the next installment of Cochrane in $CORR^{\mathbb{B}}$, our partnership between CORR[®], The Cochrane Collaboration[®], and McMaster University's Evidence-Based Orthopaedics Group. In it, researchers from McMaster University and other institutions will provide expert perspective on an abstract originally published in The Cochrane Library that we think is especially important. (Handoll HHG, Ollivere BJ, Rollins KE. Interventions for treating proximal humeral fractures in adults. Cochrane Database of Systematic Reviews 2012, Issue 12. Art. No.: CD000434. DOI: 10.1002/ 14651858.CD000434.pub3.) Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd. Reproduced with permission. Each author certifies that they, or any members of their immediate families, have no funding or commercial associations (eg, consultancies, stock ownership, equity interest, patent/licensing arrangements, etc.)

that might pose a conflict of interest in connection with the submitted article. All ICMJE Conflict of Interest Forms for authors and *Clinical Orthopaedics and* approximately 85% of all proximal humeral fractures were undisplaced, [10] but others have found proportions to be much lower [12]. Proximal humeral fractures are typical osteoporotic fractures, with women about three to four times more often affected

Y. V. Kleinlugtenbelt MD (🖂) Division of Orthopaedics, Department of Surgery, Center for Evidence-Based Orthopaedics, McMaster University, 293 Wellington St. North, Suite 110, Hamilton, ON L8L 8E7, Canada

M. Bhandari MD, PhD, FRCSC Department of Clinical Epidemiology and Biostatistics, McMaster University, Hamilton, ON, Canada

e-mail: ijdoklb@hotmail.com

than men, and older patients some 16 times more likely to experience this injury than younger patients [8]. They are the third most common fracture in elderly patients after hip and wrist fractures [4]. There is a rise in the incidence of proximal humeral fractures during the last 40 years [8, 11]. The introduction of locking-plate technology in 2002 and reverse shoulder arthroplasty in 2006 for proximal humeral fractures led to a relative increase of surgical treatment of up to 40% [1, 8], but this trend in practice was not supported by high quality evidence.

The number of displaced fracture fragments and patient age are the most important factors that influence decision-making and long-term clinical outcomes, but there is considerable variation in current clinical practice. This systematic review and meta-analysis compared surgical versus conservative treatment, different methods of surgical treatment, different methods of conservative treatment (including rehabilitation), and different methods of rehabilitation after surgical treatment for proximal humeral fractures in adults.

Related Research® editors and board members are on file with the publication and can be viewed on request. The opinions expressed are those of the writers, and do not reflect the opinion or policy of CORR[®] or The Association of Bone and Joint Surgeons[®]. Cochrane Reviews are regularly updated as new evidence emerges and in response to feedback, and The Cochrane Library (http:// www.thecochranelibrary.com) should be consulted for the most recent version of the review. This Cochrane in CORR® column refers to the abstract available at: DOI: 10.1002/ 14651858.CD000434.pub3.

Upon Closer Inspection

This meta-analysis is a comprehensive update of a prior Cochrane review that was conducted in 2001. Although seven higher quality trials were added, the major limitation of this review is still the size of the trials included, each varying from only 20 patients to 86 patients. Small clinical trials are at risk of producing misleading or erroneous findings due to low numbers of outcome events and consequent statistical fragility. Likewise, each of the included trials was a single-center study, which may pose problems for generalizability across varying populations. There was also limited or incomplete blinding in all included studies. Although patients undergoing some surgical interventions cannot always be blinded, it is often still possible to blind independent outcome assessors, healthcare providers, and data analysts.

The authors pooled the functional outcomes scores across six trials (n = 270) that compared surgical treatment versus conservative treatment, but the trials used several different functional outcome instruments. In order to

combine the data, all of the scores were converted to a standardized measure of effect called the standardized mean difference (SMD). SMDs are calculated by dividing the mean difference in scores between the two treatment groups on a particular scale by the estimated between-participants standard deviation (SD) for each trial. Unfortunately, most clinicians are not familiar with interpreting results presented in SD units, and the SDs can be under or overestimated in the presence of substantial between-study heterogeneity [7]. Authors of meta-analyses can aid the interpretation of SMD treatment effects by converting SMD scores back to a familiar scale or comparing them to a known Minimally Important Difference [7], but neither approach was incorporated in this meta-analysis.

Take-Home Messages

This Cochrane study showed no difference between surgical and conservative treatment of complex and/or proximal humeral fractures. Neither one specific method of surgical management is clearly superior to another.

There is insufficient high-quality evidence to guide the management of patients with proximal humeral fractures. Three additional small randomized controlled trials have been published since this review was updated [2, 3, 13], and recent meta-analyses suggest no benefit of surgical management in displaced three- and fourpart proximal humeral fractures in elderly [6, 9]. However, there are currently at least 14 ongoing trials, of which five are large (up to 290 patients) multicenter trials and one is a single-center trial [5] comparing surgical versus conservative treatment in displaced proximal humeral fractures. The other ongoing trials are investigating a variety of surgical approaches, implant choices, and rehabilitation regimes. These randomized controlled trials may help us arrive at stronger recommendations to help guide the management of this challenging fracture type. In anticipation of these results, broad variation in treatment preference is still justified.

Interventions for treating proximal humeral fractures in adults (Review)

Handoll HHG, Ollivere BJ, Rollins KE



This is a reprint of a Cochrane review, prepared and maintained by The Cochrane Collaboration and published in *The Cochrane Library* 2012, Issue 12

http://www.thecochranelibrary.com

WILEY

Interventions for treating proximal humeral fractures in adults (Review) Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

[Intervention Review]

Interventions for treating proximal humeral fractures in adults

Helen HG Handoll¹, Benjamin J Ollivere², Katie E Rollins³

¹Health and Social Care Institute, Teesside University, Middlesbrough, UK. ²Department of Trauma and Orthopaedics, Queen's Medical Centre, Nottingham, UK. ³Department of General Surgery, Cambridge University Hospital NHS Foundation Trust, Cambridge, UK

Contact address: Helen HG Handoll, Health and Social Care Institute, Teesside University, Middlesbrough, Tees Valley, TS1 3BA, UK. h.handoll@tees.ac.uk. H.Handoll@ete.ac.uk.

Editorial group: Cochrane Bone, Joint and Muscle Trauma Group. Publication status and date: New search for studies and content updated (conclusions changed), published in Issue 12, 2012. Review content assessed as up-to-date: 18 April 2012.

Citation: Handoll HHG, Ollivere BJ, Rollins KE. Interventions for treating proximal humeral fractures in adults. *Cochrane Database of Systematic Reviews* 2012, Issue 12. Art. No.: CD000434. DOI: 10.1002/14651858.CD000434.pub3.

Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

ABSTRACT

Background

Fractures of the proximal humerus are common injuries. The management, including surgical intervention, of these fractures varies widely. This is an update of a Cochrane review first published in 2001 and last updated in 2010.

Objectives

To review the evidence supporting the various treatment and rehabilitation interventions for proximal humeral fractures.

Search methods

We searched the Cochrane Bone, Joint and Muscle Trauma Group Specialised Register, the Cochrane Central Register of Controlled Trials, MEDLINE, EMBASE and other databases, and bibliographies of trial reports. The full search ended in January 2012.

Selection criteria

All randomised controlled trials pertinent to the management of proximal humeral fractures in adults were selected.

Data collection and analysis

Two people performed independent study selection, risk of bias assessment and data extraction. Only limited meta-analysis was performed.

Main results

Twenty-three small randomised trials with a total of 1238 participants were included. Bias in these trials could not be ruled out. Additionally there is a need for caution in interpreting the results of these small trials, which generally do not provide sufficient evidence to conclude that any non-statistically significant finding is 'evidence of no effect'.

Eight trials evaluated conservative treatment. One trial found an arm sling was generally more comfortable than a less commonly used body bandage. There was some evidence that 'immediate' physiotherapy compared with that delayed until after three weeks of immobilisation resulted in less pain and potentially better recovery in people with undisplaced or other stable fractures. Similarly, there was evidence that mobilisation at one week instead of three weeks alleviated short term pain without compromising long term outcome.

Interventions for treating proximal humeral fractures in adults (Review) Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

Two trials provided some evidence that unsupervised patients could generally achieve a satisfactory outcome when given sufficient instruction for an adequate self-directed exercise programme.

Six heterogeneous trials, involving a total of 270 participants with displaced and/or complex fractures, compared surgical versus conservative treatment. Pooled results of patient-reported functional scores at one year from three trials (153 participants) showed no statistically significant difference between the two groups (standardised mean difference -0.10, 95% CI -0.42 to 0.22; negative results favour surgery). Quality of life based on the EuroQol results scores from three trials (153 participants) showed non-statistically significant differences between the two groups at three time points up to 12 months. However, the pooled EuroQol results at two years (101 participants) from two trials run concurrently from the same centre were significantly in favour of the surgical group. There was no significant differences between the two groups in mortality (8/98 versus 5/98; RR 1.55, 95% CI 0.55 to 4.36; 4 trials). Significantly more surgical group patients had additional or secondary surgery (18/112 versus 5/111; RR 3.36, 95% CI 1.33 to 8.49; 5 trials). This is equivalent to an extra operation in one of every nine surgically treated patients.

Different methods of surgical management were tested in seven small trials. One trial comparing two types of locking plate versus a locking nail for treating two-part surgical neck fractures found some evidence of better function after plate fixation but also of a higher rate of surgically-related complications. One trial comparing a locking plate versus minimally invasive fixation with distally inserted intramedullary nails found some evidence of a short-term benefit for the nailing group. Compared with hemiarthroplasty, tension-band fixation of severe injuries using wires was associated with a higher re-operation rate in one trial. Two trials found no important differences between 'polyaxial' and 'monaxial' screws combined with locking plate fixation. One trial produced some preliminary evidence that tended to support the use of medial support locking screws in locking plate fixation. One trial found better functional results for one of two types of hemiarthroplasty.

Very limited evidence suggested similar outcomes from early versus later mobilisation after either surgical fixation (one trial) or hemiarthroplasty (one trial).

Authors' conclusions

There is insufficient evidence to inform the management of these fractures. Early physiotherapy, without immobilisation, may be sufficient for some types of undisplaced fractures. It remains unclear whether surgery, even for specific fracture types, will produce consistently better long term outcomes but it is likely to be associated with a higher risk of surgery-related complications and requirement for further surgery.

There is insufficient evidence to establish what is the best method of surgical treatment, either in terms of the use of different categories of surgical intervention (such as plate versus nail fixation, or hemiarthroplasty versus tension-wire fixation) or different methods of performing an intervention in the same category (such as different methods of plate fixation). There is insufficient evidence to say when to start mobilisation after either surgical fixation or hemiarthroplasty.

PLAIN LANGUAGE SUMMARY

Interventions for treating proximal humeral (top end of upper arm bone) fractures in adults

Fracture of the top end of the upper arm bone is a common injury in older people. It is often called a shoulder fracture. The bone typically fractures (breaks) just below the shoulder, usually after a fall. Most of these fractures occur without breaking of the skin. Often the injured arm can be supported in a sling until the fracture heals sufficiently to allow shoulder movement. More complex fractures may be treated surgically. This may involve fixing the fracture fragments together by various means. Alternatively, the top of the fractured bone may be replaced (half 'shoulder' replacement: hemiarthroplasty), or sometimes together with the joint socket (total 'shoulder' replacement). Physiotherapy is often used to help restore function.

This review includes evidence from 23 small randomised trials with a total of 1238 participants. All trials had weaknesses that could affect their results.

Eight trials evaluated conservative (non-surgical) treatment. One trial found an arm sling was generally more comfortable than a less commonly used body bandage. There was some evidence that 'immediate' physiotherapy compared with physiotherapy delayed until after three weeks of immobilisation resulted in less pain and faster recovery in people with 'stable' fractures. Similarly, there was evidence that mobilisation at one week instead of three weeks alleviated pain in the short term without compromising long term outcome. Two trials provided some evidence that patients could generally achieve a satisfactory outcome when given sufficient instruction to pursue exercises on their own.

Interventions for treating proximal humeral fractures in adults (Review) Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

Six trials, involving 270 participants with more severe fractures, compared surgical versus conservative treatment. Pooled results from three trials for patient-reported measures of function and quality of life were inconclusive. There was no difference between the two groups in mortality. However, more surgical group patients had additional or secondary surgery. This was equivalent to an extra operation in one of every nine surgically treated patients.

Seven trials tested different methods of surgical treatment. There was weak evidence of some differences (e.g. in complications) between some interventions (e.g. different devices or different ways of using devices).

There was very limited evidence suggesting similar outcomes for early versus later mobilisation after either surgical fixation or hemiarthroplasty.

Overall, there is some evidence to support earlier arm movement for less serious fractures. Otherwise, there is not enough evidence to determine the best treatment, including surgery, for these fractures.

Interventions for treating proximal humeral fractures in adults (Review) Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

References

- 1. Bell JE, Leung BC, Spratt KF, Koval KJ, Weinstein JD, Goodman DC, Tosteson AN. Trends and variation in incidence, surgical treatment, and repeat surgery of proximal humeral fractures in the elderly. *J Bone Joint Surg Am.* 2011;93:121–131.
- Buecking B, Mohr J, Bockmann B, Zettl R, Ruchholtz S. Deltoid-split or deltopectoral approaches for the treatment of displaced proximal humeral fractures? *Clin Orthop Relat Res.* 2014;472:1576–1585.
- 3. Cai M, Tao K, Yang C, Li S. Internal fixation versus shoulder hemiarthroplasty for displaced 4-part proximal humeral fractures in elderly patients. *Orthopedics*. 2012;35:e1340–1346.
- Court-Brown CM, Caesar B. Epidemiology of adult fractures: A review. *Injury*. 2006;37:691–697.
- 5. Handoll HH, Ollivere BJ, Rollins KE. Interventions for treating proximal humeral fractures in adults. *Cochrane Database Syst Rev.* 2012 Dec 12;12:CD000434.

- 6. Jia Z, Li W, Qin Y, Li H, Wang D, Zhang C, He Q, Ruan D. Operative versus nonoperative treatment for complex proximal humeral fractures: A meta-analysis of randomized controlled trials. *Orthopedics*. 2014;37: e543–551.
- Johnston BC, Patrick DL, Thorlund K, Busse JW, da Costa BR, Schunemann HJ, Guyatt GH. Patient-reported outcomes in metaanalyses-part 2: Methods for improving interpretability for decision-makers. *Health Qual Life Outcomes*. 2013;11:211.
- 8. Khatib O, Onyekwelu I, Zuckerman JD. The incidence of proximal humeral fractures in New York state from 1990 through 2010 with an emphasis on operative management in patients aged 65 years or older. *J Shoulder Elbow Surg.* 2014;23: 1356–1362.
- Mao Z, Zhang L, Zhang L, Zeng X, Chen S, Liu D, Zhou Z, Tang P. Operative versus nonoperative treatment in complex proximal humeral

fractures. *Orthopedics*. 2014;37: e410–9.

- Neer CS, 2nd. Displaced proximal humeral fractures. I. classification and evaluation. J Bone Joint Surg Am. 1970;52:1077–1089.
- 11. Palvanen M, Kannus P, Niemi S, Parkkari J. Update in the epidemiology of proximal humeral fractures. *Clin Orthop Relat Res.* 2006;442: 87–92.
- Roux A, Decroocq L, El Batti S, Bonnevialle N, Moineau G, Trojani C, Boileau P, de Peretti F. Epidemiology of proximal humerus fractures managed in a trauma center. *Orthop Traumatol Surg Res.* 2012;98:715–719.
- Sebastia-Forcada E, Cebrian-Gomez R, Lizaur-Utrilla A, Gil-Guillen V. Reverse shoulder arthroplasty versus hemiarthroplasty for acute proximal humeral fractures. A blinded, randomized, controlled, prospective study. J Shoulder Elbow Surg. 2014;23:1419–1426.